Author's response to reviews

Title: Use of a cancer registry is preferable to a direct-to-community approach for recruitment to a cohort study of wellbeing in women newly diagnosed with invasive breast cancer

Authors:

Marijana Lijovic (Marijana.Lijovic@med.monash.edu.au)
Susan R Davis (Susan.Davis@med.monash.edu.au)
Pam Fradkin (Pamela.Fradkin@med.monash.edu.au)
Maria La China (Maria.LaChina@med.monash.edu.au)
Helen Farrugia (Helen.Farrugia@cancervic.org.au)
Rory Wolfe (Rory.Wolfe@med.monash.edu.au)
Robin J Bell (Robin.Bell@med.monash.edu.au)

Version: 8 Date: 30 April 2008

Author's response to reviews: see over
30 April 2008

Dear Dr Puebla,

RE: MS 1002890308165070, Use of a cancer registry is preferable to a direct-to-community approach for recruitment to a cohort study of wellbeing in women newly diagnosed with invasive breast cancer

We were very pleased to be notified in your recent correspondence (26 April 2008) that we have received approval for the publication of our revised manuscript in *BMC Cancer*.

We thank the BMC Cancer editorial team and reviewers for their careful consideration of our manuscript.

As per the request of the editorial team, I have removed websites from within the main text and have cited them in the reference list.

Also, I have been able to resolve the reference extraction problem by referring to the Endnote advice available from the BMC website.

We trust that the manuscript will fully comply with the requirements of the journal.

Regards,

Dr Marijana Lijovic